Author's response to reviews

Title: Sleep problems and functional disability in children with functional gastrointestinal disorders: An examination of the potential mediating effects of physical and emotional symptoms

Authors:

Jennifer V Schurman (jschurman@cmh.edu)
Craig A Friesen (cfriesen@cmh.edu)
Hongying Dai (hdai@cmh.edu)
Caroline E Danda (carolinedanda@gmail.com)
Paul E Hyman (paulhyman@aol.com)
Jose T Cocjin (jtcocjin@cmh.edu)

Version: 3 Date: 12 September 2012

Author's response to reviews:

In addition to the above, I was asked to provide a more clear ethics statement, to include the actual name of the approving IRB. This revision includes that information, in addition to other reviewer requests outlined in the letter provided.